Author's response to reviews

Title: Continuing Medical Education Challenges in Chronic Fatigue Syndrome

Authors:

Dana J Brimmer (dyv4@cdc.gov)
K K McCleary (KKMcCleary@cfids.org)
Teresa A Lupton (tlupton@suddenlink.net)
Katherine M Faryna (yna@carolina.rr.com)
William C Reeves (wcr1@cdc.gov)

Version: 4 Date: 25 November 2009

Author's response to reviews:

The attached revised version addresses proofing comments from the editors.

A required major revision was to write the list of authors in the manuscript exactly as they are in the submission system (both in style and order). The second author goes by K Kimberly McCleary and the submission system listed her as K K McCleary because I did not realize that the middle name could be expanded. I have changed her name in the submission system to read K Kimberly McCleary.

The 4 minor revisions have been addressed in the attachment, specifically:

1) The title has been revised to utilize sentence case
2) Keywords have been removed from the abstract
3) All table placement text has been removed
4) Visible vertical lines have been removed from tables